Author's response to reviews

Title: Impact of insufficient drug efficacy of antiparkinson agents on patient's quality of life | A cross-sectional study

Authors:
- Jun Tsugawa (tsujun1@msn.com)
- Rieko Onozawa (riekoo@fukuoka-u.ac.jp)
- Jiro Fukae (jfukae@fukuoka-u.ac.jp)
- Takayasu Mishima (mishima1006@fukuoka-u.ac.jp)
- Shinsuke Fujioka (shinsuke@cis.fukuoka-u.ac.jp)
- Yoshio Tsuboi (tsuboi@cis.fukuoka-u.ac.jp)

Version: 7  Date: 12 June 2015

Author's response to reviews:

June 12, 2015

Editor-in-Chief
BMC Neurology

Re: Impact of insufficient drug efficacy of antiparkinson agents on patient’s quality of life

Dear, Editor

We would like to resubmit our manuscript entitled “Impact of insufficient drug efficacy of antiparkinson agents on patient’s quality of life” for consideration for publication as a original article in your Journal.

Thank you very much for your letter dated on 2015-06-6, regarding our manuscript indicated above. We wish express our appreciation to the reviewers for their insightful comments, which have helpful us significantly to improve the paper.

We have made changes on this manuscript indicated by editor.
Please find enclosed our updated manuscript with our responses point by point to editor’s comments.

I believe that the revised manuscript is suitable for publication.

Again thank you for your assistance.

Sincerely yours,
Yoshio Tsuboi, M.D, Ph.D
Director of Department of Neurology, Fukuoka University School of Medicine
E-mail: tsuboi@cis.fukuoka-u.ac.jp
Response to editor

Comment 1: Regarding consent (line 24). Can you clarify this please? Were participants made aware that returning the questionnaire indicated consent to participate in the study

Response: Thank you for editor’s comment. We added the following comments in the methods. (p. 3, line 23 )

“The questionnaire was devised specifically for use in this study.”

Comment 2: Please bring tables into the main manuscript

Response: Thank you for editor’s comment. We have added tables at the end of the main manuscript.

Comment 3: Please format references and section headings in journal style, and use initials rather than full names in Author Contributions section.

Response: Thank you for the editor’s comment. We have format references and section headings according to your journal style, and have also changed from full name to initials in Author Contribution section.

We have completed edit by native-English speaker with scientific expertise to improve the style of written English of our manuscript.